Author’s response to reviews

Title: Regulation of human chorionic gonadotropin beta subunit expression in ovarian cancer

Authors:
Aleksandra Śliwa (glodek@ump.edu.pl)
Marta Kubiczak (kubiczak@ump.edu.pl)
Anna Szczerska (abosacka@ump.edu.pl)
Grzegorz Walkowiak (grzesiek.walkowiak@gmail.com)
Ewa Nowak-Markwitz (ewamarkwitz@poczta.fm)
Beata Burczynska (b.burczynska@mdx.ac.uk)
Stephen Butler (steve.butler@mapdiagnostics.co.uk)
Ray Iles (ray.iles@mapdiagnostics.co.uk)
Piotr Białas (pbialas@ump.edu.pl)
Anna Jankowska (ajanko@ump.edu.pl)

Version: 2 Date: 15 Jul 2019

Author’s response to reviews:

Poznan, July 15th, 2019

Catherine Rice, PhD
Senior Assistant Editor
BMC Cancer

Dear Dr. Catherine Rice,

we are thankful to you as well as Dr. Anindita Chakrabarty and Dr. Priya Srinivas for reviewing our manuscript "Regulation of human chorionic gonadotropin beta subunit expression in ovarian cancer" (BCAN-D-18-02953), and for the constructive comments.
As instructed by the editor we have included a List of Abbreviations in the manuscript and added suitable information in the “Ethics approval and consent to participate” and “Funding” sections.

We hope that with the introduced amendments the revised manuscript will now be approved for the publication in the BMC Cancer.

Thank you.

Sincerely,
Aleksandra Śliwa, Ph.D.
Department of Cell Biology
University of Medical Sciences
ul. Rokietnicka 5D
60-806 Poznan, Poland